Chohan /07-Nov-05
Week 45 starting 07 Nov 05
SM18 Operation Meeting (Tuesdays)

Bench status on Monday 07 Nov 05:  11/7/2005 10:14:11 AM
	7
	A1
	MBBR3390
	Prep 5 Cool Down
	120.23

	1
	A2
	MBAL3412
	PT 12 Warm Up
	240.99

	2
	B1
	SSS261
	PT 12 Warm Up
	301.35

	5
	B2
	SSS033
	PT 6.2 Training 2
	9.19

	4
	C1
	MBBL3653
	PT 5.1 1.5 KA Quench LF
	1.91

	8
	C2
	MBBR2241
	ICS 1 WP04 HV Test Warm
	294.1

	11
	D1
	SSS218
	PT 14 Magnet MAPS
	292.4

	3
	D2
	SSS631
	SCAN 3
	1.91

	12
	E1
	MBAR1273
	PT 14 Magnet MAPS
	294.1

	9
	E2
	MBAR1255
	PT 14 Magnet MAPS
	296.88

	6
	F1
	MBAR2248
	PT 8 Fast De-excitation
	1.90

	10
	F2
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	292.4


   Tempr. from TBSM18 site
 
   
Coordinator this week: Pierre; MTM support: 16-3602

Only 15 complete Magnet Tests cycles from 12 benches 11 Dipoles (3409 Leak) + 3 SSS + 1 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 

Situation From Cryogenics:  J-P Lamboy or ACR Rep.

Special Points:
· Sunday 6/11 midday    Rocla : Not possible to change anymore magnets !
· SSSS631:  Was on ready for Chaconsa last  Fri lunchtime but no one to do it ( Marco ?) ; How much more  time for MM this week? 
AOB 
	Magnets under Test from 30-10-2005 to 05-11-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for

	1
	TBA1
	MBBL3409(2) 
	[image: image3.png]



	26-10-2005 
	32 h.
	20 h.
	21 h.
	16 h.
	14 h.
	01-11-2005 
	103 h.
	Stand By

	2
	TBA1
	MBBL3409(2) 
	[image: image4.png]



	01-11-2005 
	16 h.
	22 h.
	9 h.
	21 h.
	12 h.
	05-11-2005 
	80 h.
	Stripping

	3
	TBA2
	MBBR3340 
	[image: image5.png]



	29-10-2005 
	32 h.
	24 h.
	43 h.
	16 h.
	14 h.
	04-11-2005 
	129 h.
	Stripping

	4
	TBB1
	SSS653 
	[image: image6.png]



	28-10-2005 
	87 h.
	14 h.
	25 h.
	17 h.
	25 h.
	04-11-2005 
	168 h.
	Stand By

	5
	TBB2
	SSS113 
	[image: image7.png]



	27-10-2005 
	28 h.
	13 h.
	17 h.
	8 h.
	9 h.
	30-10-2005 
	75 h.
	Stripping

	6
	TBB2
	SSS166 
	[image: image8.png]



	30-10-2005 
	35 h.
	12 h.
	37 h.
	17 h.
	14 h.
	04-11-2005 
	115 h.
	Stripping

	7
	TBC1
	MBAL3139(2) 
	[image: image9.png]



	24-10-2005 
	27 h.
	21 h.
	25 h.
	61 h.
	18 h.
	30-10-2005 
	152 h.
	Stripping

	8
	TBC1
	MBBL3407 
	[image: image10.png]



	30-10-2005 
	35 h.
	32 h.
	16 h.
	15 h.
	13 h.
	04-11-2005 
	111 h.
	Stripping

	9
	TBC2
	MBBR3415 
	[image: image11.png]



	26-10-2005 
	39 h.
	25 h.
	17 h.
	15 h.
	17 h.
	31-10-2005 
	113 h.
	Stripping

	10
	TBD1
	SSS272 
	[image: image12.png]



	28-10-2005 
	40 h.
	11 h.
	16 h.
	9 h.
	15 h.
	01-11-2005 
	91 h.
	Stripping

	11
	TBE1
	MBAR1270 
	[image: image13.png]



	29-10-2005 
	23 h.
	22 h.
	14 h.
	12 h.
	21 h.
	02-11-2005 
	92 h.
	Stripping

	12
	TBE2
	MBAL1072 
	[image: image14.png]



	27-10-2005 
	36 h.
	21 h.
	13 h.
	17 h.
	24 h.
	01-11-2005 
	111 h.
	Stripping

	13
	TBF1
	MBAL2209 
	[image: image15.png]



	25-10-2005 
	39 h.
	25 h.
	18 h.
	15 h.
	13 h.
	30-10-2005 
	110 h.
	Stripping

	14
	TBF1
	MBAR1264 
	[image: image16.png]



	30-10-2005 
	35 h.
	26 h.
	23 h.
	20 h.
	16 h.
	04-11-2005 
	120 h.
	Stripping

	15
	TBF2
	MBAL3395 
	[image: image17.png]



	27-10-2005 
	22 h.
	35 h.
	10 h.
	15 h.
	35 h.
	01-11-2005 
	117 h.
	Stripping

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10

	1
	MBBL3409
	12995.2 A.
(D1-U)
	12840.4 A.
(D2_L)
	12850 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3409
	11524.4 A.
(D2_U)
	12850 A.
[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR3340
	10336.4 A.
(D2_L)
	12068.5 A.
(D2_L)
	12390.6 A.
(D1-U-S01)
	12684.2 A.
(D2_U_S13)
	12850 A.
[image: image20.jpg]



	-
	-
	-
	-
	-

	4
	SSS653
	5434 A.
(MQM_E)
	5674 A.
(MQM_E)
	5820 A.
[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-

	5
	SSS113
	10379.8 A.
(QE_12)
	12358.3 A.
(QE_12)
	12657.3 A.
(QE_34)
	[image: image22.png]



	-
	-
	-
	-
	-
	-

	6
	SSS166
	11672.1 A.
(QE-12)
	12340.8 A.
(QI-34)
	12729.1 A.
(QI_34)
	[image: image23.png]



	-
	-
	-
	-
	-
	-

	7
	MBAL3139
	10609 A.
(D2_L)
	11496 A.
(D1-U)
	12277 A.
(D2_L)
	12376 A.
(D1_L)
	12833 A.
(D1-L)
	[image: image24.png]



	-
	-
	-
	-

	8
	MBBL3407
	11853.7 A.
(D2_L)
	12123.9 A.
(D2_U)
	12559.7 A.
(D2_U)
	[image: image25.png]



	-
	-
	-
	-
	-
	-

	9
	MBBR3415
	12402 A.
(D2_L)
	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS272
	11296.7 A.
(QE-12)
	12581.7 A.
(QE_34)
	12734 A.
(QE-34)
	[image: image27.png]



	-
	-
	-
	-
	-
	-

	11
	MBAR1270
	11809.3 A.
(D1_L)
	12367 A.
(D2-U)
	12839.4 A.
(D2-L)
	[image: image28.png]



	-
	-
	-
	-
	-
	-

	12
	MBAL1072
	11427.2 A.
(D2_L)
	12741.5 A.
(D1_U)
	12850 A.
[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-

	13
	MBAL2209
	11247.7 A.
(D1_U)
	12532.6 A.
(D1-U)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-

	14
	MBAR1264
	11415 A.
(D1-L)
	11894 A.
(D1-U)
	12321.6 A.
(D2-L)
	12531.6 A.
(D2-L)
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	-
	-
	-
	-
	-

	15
	MBAL3395
	12139.2 A.
(D1_L)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
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Long Connect time for SSS653 ;   Super good Noell Magnets at the end of their Series (3395, 3409,3415  :1 quench Ultimate )
3409 behaviour in 2nd run due to leak rather strange : dropped down  for 1st quench  to 11500 A after Ultimate reached in 1st run & the second run was  ON the BENCH  with warm up disconnection & Connection

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Thursday_01_A
	TBA2


	Repair of the pumping valve of the CFB
	ACR
	
	No power test on test bench during the afternoon.

	Tuesday_01_N
	All


	Cryo stop. All cold tests stopped.
	ACR
	
	General stop of the liquefactor. Stop 3 hours

	Wednesday_02_A
	TBC1


	No cold test can be launched because PLCSE was blocked.
	ACR
	
	Hardware reset done the OK.

	Wednesday_02_N
	TBC1, TBF1


	File not present in storage for C1 MBBL3407 .ka01 and F1 MBAR1264 .a03.
	AB/CO
	
	B Khomenko has been informed for correction.

	Thursday_03_N
	TBC2


	No signal using hfoff for Vcathode and Vanode on MB3407 during diode test.
	MTM
	hf
	Signal found OK with lf off.

	Friday_04_A
	TBC2


	Problem in generating config.file for warm-up quench
	MTM
	
	Equip.support contacted.

	Friday_04_A
	TBC1


	MB3653 installed. No Temperature sensor attached in MTF
	CRI
	
	NC opened. HWREC put off.

	Friday_04_A
	TBA2


	MB2244: Quench during SPA test in the ramp down at 3700 A.
	AB/PO?
	
	Equip.support told to repeat the test.

	Friday_04_A
	TBE2


	HV CDBP on MB1255: test failed on Temperature sensor vs ground (5 A) instead of 2 A. The sensor still is working.
	MTM
	
	Equip.support checked the results and advised to open a NC. Test went on with Tsensor connected to have the recording with lf archiver during the run at cold.

	Friday_04_A
	TBC2


	MB2231: No diode bump observed during PTE. 
	MTM
	
	Equip.support found that the cables to record the diode signals were not connected. He asked to repeat the test. Signal was then observed.

	Friday_04_N
	TBF2


	MB2244: Spike and noise seen in trigger & current signals of final warm-up quench.
	MTM
	
	Equipment support and TC contacted. TC asked to repeat the warm-up quench. 



	Friday_04_N
	TBB1


	Triggers on potaim cards during IAP at warm on SSS261.
	MTM
	
	ICS has to check the connections.


	Saturday_05_M
	TBF1


	TEMA application blocked during IAP at warm
	AB/CO
	
	Crates reset. Then OK.


	Saturday_05_M
	TBE1


	SPA occurred during de-ramp of PT8 at around 5000A. 

Same during PT10.
	MTM
	
	D20 connector was found disconnected.

	Sunday_06_A
	TBB1


	SS261: On launching SPA test at 1000 A, the power converter is tripping when the current starts to ramp up from 50A.
	AB/PO
	
	Equip.support checked the faults and reset the power converter.


	Sunday_06_A
	TBC2


	Connection of the MB2241 cannot go on because magnet is not positioned properly.
	ICS
	
	ROCLA team did not answer to phone calls. Connection stopped up to Monday.


	Sunday_06_A
	TBC2
	Equip.support came to put HWREC on for tests of MB2241 (put off for the cold tests of MB3653)
	
	
	

	Sunday_06_A
	TBA2
	After warm up quench on MB3412 OP team signed the magnet on stand_by.  Cryo team did not notice the message in the e-traveller and cool down the magnet again.
	ACR
	
	Warm up quench repeteated.


